Author’s response to reviews

Title: Prune Belly Syndrome and dental anomaly association: a case report.

Authors:

Maria D Basso (mdanibs@gmail.com)
Carla O Favretto (carla.favretto@gmail.com)
Robson F Cunha (cunha@foa.unesp.br)

Version: 2 Date: 10 September 2012

Author’s response to reviews:

Dear Editor BMC,

We modify, as request, the consent to the publication in the text of the manuscript, changing the word patient for parents who have consented.

We also sent an email with the consent for publication signed by the mother of the patient.

Thanks

Prof. Robson Frederico Cunha